## **Author's response to reviews**

**Title:** Environment And Genetics in Lung cancer Etiology (EAGLE) study: a novel population-based integrative case-control study of lung cancer

## Authors:

Maria Teresa Landi (landim@mail.nih.gov)

Dario Consonni (dario.consonni@unimi.it)

Melissa Rotunno (rotunnom@mail.nih.gov)

Andrew W Bergen (bergena@mail.nih.gov)

Alisa Goldstein (goldstea@mail.nih.gov)

Jay Lubin (<a href="mailto:lubinj@mail.nih.gov">lubin (lubinj@mail.nih.gov</a>)

Lynn Goldin (goldinl@mail.nih.gov)

Michael Alavanja (alavanjam@mail.nih.gov)

Glen Morgan (gmorgan@mail.nih.gov)

Amy Subar (<a href="mailto:subara@mail.nih.gov">subar (<a href="mailto:subara@mail.nih.gov">subara@mail.nih.gov</a>)

Fabrizio Previdi (fprevidi@craon.it)

Massimo Corno (mcorno@impresaedilemariani.it)

Maurizia Rubagotti (maurizia.rubagotti@unimi.it)

Barbara Marinelli (barbara.marinelli@unimi.it)

Benedetta Albetti (benedetta.albetti@unimi.it)

Antonio Colombi (antonio.colombi@unimi.it)

Margaret Tucker (tucker@mail.nih.gov)

Wacholder Sholom (wacholders@mail.nih.gov)

Angela C Pesatori (angela.pesatori@unimi.it)

Neil E Caporaso (caporasn@mail.nih.gov)

Pier Alberto Bertazzi (pieralberto.bertazzi@unimi.it)

Version: 2 Date: 25 October 2007

## **Author's response to reviews:**

I am re-submitting both cover letter and manuscript because the link you provided

(http://www.biomedcentral.com/imedia/1462024222165547\_comment.pdf=20

=20) results empty and I am not sure the cover letter and manuscript have been uploaded correctly.

Please see below the cover letter:

Dear Editor.

We are submitting the manuscript ¿Environment And Genetics in Lung cancer Etiology (EAGLE) study: a novel population-based integrative case-control study of lung cancer; for your consideration for online publication to BMC Public Health.

We describe a novel large case-control study of lung cancer and smoking - two conditions of unparalleled public health importance. The study (EAGLE) has a novel integrative design. We describe here details on design, population, infrastructure, database, biospecimen resource and early findings from EAGLE.

As a number of EAGLE-based reports will emerge over the next few months, we expect this manuscript will be widely cited. In particular, at the beginning of next year, results from a genome-wide association study of lung cancer and smoking within EAGLE will emerge and genotype and selected phenotype information will be made available to qualified investigators. Thus, it is crucial for us to be able to refer all interested investigators to find details on the EAGLE study on an open journal with rapid access.

In summary, we are confident this will be a highly informative and timely manuscript for BMC readership. Given the broad interest of the topics addressed by this manuscript, we would welcome a routing of the manuscript to BMC Medicine, should you feel it appropriate.

This manuscript is not under consideration by any other journal. All authors have read it and agreed with its content. The research reported in the manuscript has been performed with the approval of an appropriate ethics committee and has been carried out in compliance with the Helsinki Declaration. Authors declare no conflict of interest.

Appropriate reviewers with expertise in lung cancer and tobacco issues relevant to public health include:

Chris Amos MD Anderson
David Christiani Harvard
Margaret Spitz MD Anderson
Thomas Sellers Moffitt cancer center

Sincerely,

Maria Teresa Landi

Maria Teresa Landi, M.D., Ph.D.

Senior Investigator

Genetic Epidemiology Branch

Division of Cancer Epidemiology and Genetics

National Cancer Institute, NIH

6120 Executive Blvd., EPS 7114

Bethesda, MD 20892-7236

Phone: 301-402-9519

Fax: 301-402-4489

landim@mail.nih.gov